Author's response to reviews

Title: Methylenetetrahydrofolate Reductase (MTHFR) C677T polymorphism and high plasma homocysteine in Chronic Hepatitis C (CHC) infected patients from the Northeast of Brazil.

Authors:

Erika Rabelo F Siqueira (erikarfs@hotmail.com)
Cláudia PM Souza Oliveira (cpm@usp.br)
Maria Tereza C Muniz (tcartaxo.upe@gmail.com)
Filipe S Silva (filipe.santana.silva@gmail.com)
Leila MMB Pereira (leilapereira@pq.cnpq.br)
Flair J Carrilho (fjcarril@usp.br)

Version: 3 Date: 29 July 2011

Author's response to reviews: see over
Dear Prof.

We are resubmitting the Manuscript: 1601730666548107-R1 entitled: “Methylenetetrahydrofolate Reductase (MTHFR) C677T polymorphism and high plasma homocysteine in Chronic Hepatitis C (CHC) infected patients from the Northeast of Brazil”, by Siqueira ERF, Oliveira CPMS, Muniz MTC, Silva F, Pereira LMMB, Carrilho FJ. The manuscript was corrected according to the suggestions of the reviewers. Please, find below our considerations to the reviewer’s comments.

**Major revisions**

1) Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs).

   **Comments**
   Done

2) Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution.

   **Comments**
   Done

3) References: All references should be cited in the text and should be cited using consecutive numbers in square brackets. Please do not use superscript numbers. Also, please place the reference before the full stop. For example: text [1]. Also reference 44 does not appear to be cited.

   **Comments**
   Corrected the mistake
4) 'Contributions Section': Please rename this section so it is now called Author Contributions and please confirm that all authors read and approved the final manuscript.
   Comments
   Renamed the section

5) Tables: A total of eight tables are given at the end of the manuscript but only six are cited throughout the manuscript. Please ensure that all tables are cited and they are cited in numerical order.
   Comments
   Corrected the mistake

Minor revisions

1) Highlighting: Please remove all forms of highlighting present in the manuscript.
   Comments
   Done

2) Comments: Throughout the manuscript comments such as the one after the abstract section heading on page 2 '(Corrections- Reviewer 2- question 3)'. Please remove these.
   Comments
   Done

3) Abbreviations: Please put this section into paragraph form.
   Comments
   Done

4) Tables: Please remove any visible vertical lines present in the manuscript.
   Comments
   We had some difficulties, please check if it are in the correct form.

5) Additional Files: Please remove the attached additional file as this is no longer needed.
   Comments
   Done
6) Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Comments
Done

Thank you again for your kind consideration of the revised manuscript.

Best regards,

Claudia PMS Oliveira, MD. PhD
Department of Gastroenterology
University of Sao Paulo School of Medicine
Av. Dr. Enéas de Carvalho Aguiar nº 255, Instituto Central, # 9159.
05403-000 - Sao Paulo - Brazil
Phone: 55-11-3069-6447
Fax: 55-11-3069-7830
E-mail: cpm@usp.br